Author's response to reviews

Title: Clinical Implication of ZEB-1 and E-cadherin Expression in Hepatocellular carcinoma

Authors:

Motoyuki Hashiguchi (moto-h@m2.kufm.kagoshima-u.ac.jp)
Shinichi Ueno (ueno1@m.kufm.kagoshima-u.ac.jp)
Masahiko Sakoda (sakoda@m.kufm.kagoshima-u.ac.jp)
Satoshi Iino (iino@m2.kufm.kagoshima-u.ac.jp)
Kiyokazu Hiwatashi (ino@m2.kufm.kagoshima-u.ac.jp)
Koji Minami (k373-as@m.kufm.kagoshima-u.ac.jp)
Kei Andou (andou@m2.kufm.kagoshima-u.ac.jp)
Yuko Mataki (mataki@m.kufm.kagoshima-u.ac.jp)
Kosei Maemura (kmaemura@m3.kufm.kagoshima-u.ac.jp)
Hiroyuki Shinchi (shinchi@m.kufm.kagoshima-u.ac.jp)
Sumiya Ishigami (igeka@m2.kufm.kagoshima-u.ac.jp)
Syoji Natsugoe (natsugoe@m2.kufm.kagoshima-u.ac.jp)

Version: 3 Date: 6 November 2013

Author's response to reviews: see over
To Editor-in-Chief of BMC cancer:

Thank you for your reply that accepted my article.

We are sending the manuscript revised partially, because you recommended that Competing interests, Authors’ contributions and Acknowledgement should be added.

The attached paper titled "Clinical Implication of ZEB1 and E-cadherin Expression in Hepatocellular carcinoma" has been carefully reviewed by an experienced medical editor whose first language is English and who specializes in the editing of papers written by physicians and scientists whose native language is not English. We attached a certificate of the editing.

We thank for your consideration and look forward to hearing from you. If you have any questions concerning the manuscript, we will be happy to answer them.

Sincerely yours

Motoyuki Hashiguchi
Department of Digestive Surgery, Breast and Thyroid Surgery
Kagoshima University Graduate School of Medical sciences
8-35-1, Sakuragaoka, Kagoshima, 890-8520, Japan
moto-h@m2.kufm.kagoshima-u.ac.jp